Spatial Distribution and Risk Factors of Highly Pathogenic Avian Influenza (HPAI) H5N1 in China

Vincent Martin¹, Dirk U. Pfeiffer², Xiaoyan Zhou¹, Xiangming Xiao³, Diann J. Prosser^{4,5}, Fusheng Guo¹, Marius Gilbert^{6,7}*

1 Emergency Centre for the Control of Transboundary Animal Diseases, Food and Agriculture Organization of the United Nations (FAO), Beijing, China, 2 Veterinary Epidemiology & Public Health Group, Department of Veterinary Clinical Sciences, The Royal Veterinary College, University of London, London, United Kingdom, 3 Department of Botany and Microbiology, Center for Spatial Analysis, University of Oklahoma, Norman, Oklahoma, United States of America, 4 USGS Patuxent Wildlife Research Center, Beltsville, Maryland, United States of America, 5 University of Maryland, College Park, Maryland, United States of America, 6 Biological Control and Spatial Ecology, Université Libre de Bruxelles, Brussels, Belgium, 7 Fonds National de la Recherche Scientifique, Brussels, Belgium

Abstract

Highly pathogenic avian influenza (HPAI) H5N1 was first encountered in 1996 in Guangdong province (China) and started spreading throughout Asia and the western Palearctic in 2004–2006. Compared to several other countries where the HPAI H5N1 distribution has been studied in some detail, little is known about the environmental correlates of the HPAI H5N1 distribution in China. HPAI H5N1 clinical disease outbreaks, and HPAI virus (HPAIV) H5N1 isolated from active risk-based surveillance sampling of domestic poultry (referred to as HPAIV H5N1 surveillance positives in this manuscript) were modeled separately using seven risk variables: chicken, domestic waterfowl population density, proportion of land covered by rice or surface water, cropping intensity, elevation, and human population density. We used bootstrapped logistic regression and boosted regression trees (BRT) with cross-validation to identify the weight of each variable, to assess the predictive power of the models, and to map the distribution of HPAI H5N1 risk. HPAI H5N1 clinical disease outbreak occurrence in domestic poultry was mainly associated with chicken density, human population density, and elevation. In contrast, HPAIV H5N1 infection identified by risk-based surveillance was associated with domestic waterfowl density, human population density, and the proportion of land covered by surface water. Both models had a high explanatory power (mean AUC ranging from 0.864 to 0.967). The map of HPAIV H5N1 risk distribution based on active surveillance data emphasized areas south of the Yangtze River, while the distribution of reported outbreak risk extended further North, where the density of poultry and humans is higher. We quantified the statistical association between HPAI H5N1 outbreak, HPAIV distribution and post-vaccination levels of seropositivity (percentage of effective post-vaccination seroconversion in vaccinated birds) and found that provinces with either outbreaks or HPAIV H5N1 surveillance positives in 2007–2009 appeared to have had lower antibody response to vaccination. The distribution of HPAI H5N1 risk in China appears more limited geographically than previously assessed, offering prospects for better targeted surveillance and control interventions.

Citation: Martin V, Pfeiffer DU, Zhou X, Xiao X, Prosser DJ, et al. (2011) Spatial Distribution and Risk Factors of Highly Pathogenic Avian Influenza (HPAI) H5N1 in China. PLoS Pathog 7(3): e1001308. doi:10.1371/journal.ppat.1001308

Editor: Neil M. Ferguson, Imperial College London, United Kingdom

Received August 3, 2010; Accepted January 31, 2011; Published March 3, 2011

This is an open-access article distributed under the terms of the Creative Commons Public Domain declaration which stipulates that, once placed in the public domain, this work may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose.

Funding: This study was supported by the Food and Agriculture Organization of the United Nations and funded by the United States Agency for International Development (USAID) through the FAO Emergency Center for Transboundary Animal Diseases (ECTAD) office in China. The study was also supported by the National Institutes of Health Fogarty International Center through the NSF/NIH Ecology of Infectious Diseases program (7R01TW007869-04). Dr. Xiao is also supported by the Chinese Special Program for Prevention and Control of Infectious Diseases (No. 2008ZX10004-012) from the China Ministry of Health and or role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: mgilbert@ulb.ac.be

Introduction

HPAI H5N1 virus infection was first encountered in China in 1996 in the southern part of the country with the discovery of a virus that killed geese in Guangdong province (Goose/GD/96) [1]. In 1997, Hong-Kong experienced the first major outbreak of HPAI H5N1 associated with several human deaths, alerting the international community to the potential threat caused by this new strain of HPAI virus (HPAIV). Between 1999 and 2003, the virus underwent a series of evolutionary changes and multiple genotypes of HPAIV H5N1 detected through routine live bird market surveillance in southern China emerged, indicating that the virus was still active and widely circulating [2]. However, the first major outbreak of HPAI H5N1 in mainland China started in January 2004 in Guangxi autonomous region, in southern China, bordering Vietnam. As the outbreak unfolded, the disease was detected widely throughout the country, causing over 110 outbreaks in 23 provinces since the onset of the epidemic and leading to the culling of more than 35 million poultry to curb the spread of the disease.

To answer the challenge of controlling HPAI H5N1 across such a vast territory characterized by a diversity of agricultural production systems and economic development, China has taken several important steps to confront and control outbreaks and deal with the occurrence of human cases. These steps include measures such as stamping out, movement controls, cleaning and disinfection of infected premises, and the adoption of a nationwide massive vaccination campaign combined with intensive postvaccination surveillance efforts. Effective vaccines have been developed and disease outbreaks have been responded to in a

Author Summary

The geographical distribution of highly pathogenic avian influenza (HPAI) H5N1 and agro-ecological risk factors have been studied in a number of countries in Southeast Asia. However, little is know of its distribution in China where HPAI H5N1 first emerged in 1996, evolved, and spread throughout Asia and the western Palearctic in 2004–2006. This study analyzes separately the distribution, in domestic poultry, of HPAI virus (HPAIV) H5N1 isolated from active risk-based surveillance sampling and HPAI H5N1 clinical disease outbreaks. These data are analyzed in relation to the distribution of chicken and domestic waterfowl population density, proportion of land covered by rice or surface water, cropping intensity, elevation, and human population density. HPAI H5N1 viruses identified by risk-based surveillance are found to be associated with domestic waterfowl density, human population density, and the proportion of land covered by surface water. In contrast, HPAI H5N1 clinical disease outbreak occurrences were mainly associated with chicken density, human population density, and low elevation. These results show that the distribution of HPAI H5N1 risk in China appears more limited geographically than previously assessed, offering prospects for better targeted surveillance and control interventions.

timely and effective manner, improving China's capacity to contain the disease and drastically reducing the number of outbreaks over the past years, with no outbreak detected in 2010. The year 2005 represented a turning point in the control strategy with the enforcement of a so-called universal vaccination campaign, when vaccination became compulsory for all poultry using the H5N1 Re-1 (A/Goose/Guangdong/1/96-PR8) vaccine strain. In parallel, a large amount of surveillance testing has been conducted both at provincial and national levels with the collection of an average of 4.7 million samples every year during the period 2007-2009 for the detection of silent viral circulation, the possible emergence of new strains and ultimately the identification of potential vaccination failure. Through the national surveillance program for the monitoring of HPAIV H5N1 circulation, the virus has been regularly detected, generating essential information for understanding the infection distribution pattern. More specifically, it has provided evidence that, despite a reduction in reported HPAI H5N1 outbreaks, some parts of the country still offer a favorable breeding ground for influenza viruses to circulate and potentially novel strains to emerge, representing a threat for the generation of new influenza pandemic strains. This particularly applies to the southern part of the country, which has historically been referred to as a hypothetical influenza epicenter [3] where agricultural and cultural practices place man and animals in close proximity. However, very few studies have actually attempted to map the potential distribution of avian influenza disease and infection risk across the diversity of ecological, cultural and production systems present in China. This lack of a sound description of HPAIV H5N1 geographical niches makes it difficult to refine control strategies that rely heavily on vaccination and that would greatly benefit from more targeted risk management. Better understanding of the infection dynamic pattern, the environmental and ecological factors associated with persistence of the disease in various poultry production systems will significantly strengthen efforts to achieve disease control and exclude infection from major poultry production centers, thus optimizing resources allocated to controlling the disease and reducing the risk for human infection.

This study aimed to analyze the interrelationship of HPAI H5N1 in China with its environment, by exploring the association between selected spatial risk factors and two different indicators of HPAIV H5N1 presence, namely reported clinical outbreak occurrence in poultry and detection of sub-clinical HPAIV infection through risk-based surveillance. The study benefits from several improvements over previous work [4]. First, the analysis is not only based on HPAI H5N1 outbreak data which are of limited value in a context of massive vaccination (especially after the implementation of the national vaccination campaign which began towards the end of 2005), but also uses the results of HPAI H5N1 monitoring implemented as part of the national active surveillance program in live bird markets from 2007 to 2009, termed risk-based surveillance in the rest of this manuscript. We also make use of an estimation of vaccine efficacy as measured by the Haemagglutination Inhibition (HI) test of serological samples collected monthly from poultry at province level. Second, we use updated poultry census data that differentiates between chicken, ducks and geese. This distinction is important as shown by previous HPAI H5N1 disease mapping efforts [5,6], and is probably associated with differences in susceptibility to HPAI H5N1 virus between these species. Third, we used and compared the outputs of bootstrapped logistic regression and boosted regression trees with cross-validation, so that we could robustly estimate the weight of each tested risk factor, the goodness of fit of our predictions, and to allow us to map both the prediction of risk as well as its uncertainty.

Methods

Data

Two types of data relating to HPAI H5N1 presence have been used as dependent variables in this study.

First, poultry HPAI H5N1 disease outbreak data were compiled from two main sources: (1) one being the Official Veterinary bulletin published on the website of the Chinese Ministry of Agriculture (MoA; http://www.agri.gov.cn/); and (2) the other source coming from official reports to the World Organisation for Animal Health (OIE) that were compared with MoA's report. Where there was an inconsistency in the outbreak date or location, we obtained accurate data through web research and consultation of local experts. Ninetyfive percent of poultry outbreak data had detailed address information which was then geocoded. The remaining 5% for which no accurate location could be obtained were geocoded using the prefecture centroid (administrative level 4).

Second, the Ministry of Agriculture in China routinely coordinates a surveillance program twice a year at the national level and monthly at provincial level in live bird markets consisting of sampling domestic poultry for the detection of HPAIV H5N1. The selection of markets is based on their characteristics with regard to size, trade and hygiene practices which are assumed to increase the likelihood of detecting the virus. All samples collected at provincial level are tested by polymerase chain reaction (PCR). All AI positive samples are sent to the Harbin National Veterinary Research Institute for confirmation, subtyping and virus isolation. The positive HPAIV H5N1 findings are then reported at the central government level and the data are released by the Veterinary Bureau in the MOA through the monthly Official Veterinary Bulletin, from which we extracted data on positive identification of HPAIV H5N1 between January 2007 and September 2009. The surveillance data were geocoded using the market location when the market name was available. For 15% of the data, however, this information was missing and positives were geocoded using the prefecture centroid.

In addition to HPAI H5N1 positives, the monthly proportion of post-vaccination seropositive samples was also extracted from the Monthly Official Veterinary Bulletin at the province level. Postvaccination monitoring is performed on a regular basis at provincial level to assess the efficacy of the vaccination. Chickens, ducks and geese are sampled 21 days post-immunisation and an effective immune response is defined as a sero-conversion in bird with titres >4Log₂ when measured by HI test, using homologous antigen, similar to the vaccine strain. Similar to the surveillance results, post-vaccination serological results obtained are collected at national level and published in the Official Veterinary Bulletin.

The spatial distribution of HPAI H5N1 was investigated using a set of 7 explanatory variables which are known to be important risk factors, based on published scientific evidence and expert opinion. First, we considered the abundance of chickens, and domestic waterfowl separately based on previous work that had demonstrated a weak positive association between HPAI H5N1 presence and chicken density [6,7], but a stronger association with duck density [5]. Second, anthropogenic variables were found to be associated with HPAI H5N1 in a number of studies conducted in countries with very different agro-ecological conditions such as Thailand, Bangladesh, Vietnam, and Romania [6,8,7,9,10], and we therefore chose to include human population density. Third, several studies also identified land use and cropping variables as significant predictors of HPAI H5N1 presence in Asia. For example, Pfeiffer et al. [7] found HPAI H5N1 to be associated with the proportion of land occupied by aquaculture and by rice paddy fields in Vietnam, and Gilbert et al. [6] found a strong association with rice cropping intensity in Thailand. Similarly, statistically significant effects of access to water, or density of waterways were identified by Biswas et al [11] for Bangladesh and Ward et al. [12] for Romania. Therefore, we decided to include three variables: the proportion of land occupied by water (running water or water bodies), the proportion of land occupied by rice paddy fields, and the cropping intensity (number of crops cultivated in an unit area of cropland over a year). Finally, we included elevation in our analysis since several studies have reported an increased HPAI H5N1 risk in lowland and river delta areas [6,7,13]. The risk factor variables and corresponding data sources are presented in Table 1.

The analysis was carried out at a spatial resolution of 0.0833 decimal degrees of latitude and longitude (approximately 5.5 to 8.8 km for the study area comprised between 54 and 18 degrees of latitude north). The statistical methods used in this study required having risk factor variable values for a large set of locations where HPAIV H5N1 or HPAI H5N1 outbreaks would be considered absent (negatives), so as to contrast the agro-ecological conditions

Table	1.	Risk	factor	variable	s used	in	the	anal	ysis.
-------	----	------	--------	----------	--------	----	-----	------	-------

Abbreviation	Description	Transform	Source
ChDnLg	Chickens/km ²	Log10[x+1]	Robinson et al. [42]
DuGeDnLg	Domestic waterfowls/km ²	Log10[x+1]	Robinson et al. [42]
HpDnLg	People/km ²	Log10[x+1]	GRUMP [43]
WaPc	% land occupied by water	-	Jiyuan et al. [44]
RiPc	% land occupied by rice crop	-	Jiyuan et al. [44]
Cint	Average cropping intensity	-	Xiao et al. [45]
DemLg	Mean Elevation	Log10[x+1]	GTOPO30 [46]

doi:10.1371/journal.ppat.1001308.t001

associated with HPAIV H5N1 or HPAI H5N1 outbreaks presence (positives). Negatives were selected randomly from throughout the country based on three conditions: i) no HPAI H5N1 outbreaks had been reported and no HPAIV H5N1 positive results had been obtained from the active risk-based surveillance; ii) being at a minimum distance >0.0833 decimal degree of any positive; and iii) being in a location where human population density was >1 person/km² to exclude desert and high mountain areas from the analysis since the focus of this analysis was on locations with likely relevance for disease maintenance in poultry.

Analysis

Two approaches were used to model the spatial distribution of HPAI H5N1 presence or absence: multiple logistic regression and boosted regression trees (BRT). Logistic regression allows predicting a variable with a binary response, such as the presence or absence of a disease, as a function of a number of variables, or predictors. Logistic regression models have been used in a number of studies trying to identify environmental correlates and risk factors associated with HPAI H5N1 presence [6,7,10]. However, one limitation of logistic regression is the necessity to perform specific adjustments to accommodate non-linearity of effect of the continuous-scale risk factors on the logit form of the outcome variable. Two approaches have been described to account for this. First, a risk factor can be added to the model as a quadratic term so that predicted probabilities of presence can be maximum (or minimum) for intermediate values (e.g. [6,9]). Second, each continuous-scale risk factor variable is converted into a nominalscale variable where each category level represents a particular range of values in the original variable (e.g. [14,7,10]). However, both methods have their limitations. The first method can only partially model more complex non-linear dependencies, and the second method is sensitive to the range of values represented by each category level. In the presence of spatial autocorrelation, logistic regression requires the use of relatively complex estimation algorithms. BRT has been developed relatively recently for predicting the distribution of organisms [15]. It is very efficient for dealing with non-linear relationships and interactions between variables. It can be considered a disadvantage that it does not have the facility to assess the statistical significance of individual effect variables, though it allows estimating the relative importance of each variable to the predictions. In a comprehensive review of presence/absence distribution modeling methods, Elith et al. [16] found BRT to perform best along with the maximum entropy method. Elith et al. [15] published a detailed description of an analysis approach using BRT which implements a cross-validation procedure allowing identification of model parameters. In this study, we compare logistic regression and BRT in terms of validity and ease of interpretation of the outputs. We also discuss our findings in relation to the spatial patterns of HPAI H5N1 described in previously published work that used logistic regression methods

In the logistic regression method, all variables were forced in the model, and the likelihood ratio test was used to assess the contribution of each variable to the predictions. For the BRT model, we used 10 sets of training and test points for cross-validation, a tree complexity of 4, a learning rate of 0.005 and a bag fraction of 75%. Using those parameters, the cross-validation stepwise function presented by Elith et al. [15] was used to identify the optimal number of trees in the model. The weight of each variable estimated over the identified number of trees was used as an indicator of each variable's importance for predicting HPAI H5N1 presence/absence. One should note that those weights are not absolute metrics, and the weights of all variables of a BRT

model sum to 1. This analysis was conducted using two outcome variables, first HPAI H5N1 outbreak occurrence during the entire study period and second HPAIV H5N1 positive findings between 2007 and 2009.

Assessing the performance of our models directly from the logistic regression and BRT predicted probabilities and observed presence/ absence had two main limitations. First, logistic regression performance metrics have been shown to be sensitive to low (<10%) and high (>90%) frequencies of the binary outcome categories [17]. The proportion of positives in our dataset was extremely low, and it was therefore necessary to address this potential bias (to our knowledge, the presence of this potential bias has not been thoroughly assessed for BRT models, but see [18]). Second, quantifying model performances using the data set used to train the model tends to inflate the performance metrics compared to a situation where an independent data set is used. We developed a bootstrapping procedure aiming to generate a robust estimate of model performance by simultaneously addressing those two limitations.

The bootstrapping analysis involved a series of sequential steps: (i) selection of a balanced subset of data from the complete dataset: all *n* points with HPAI H5N1 presence were included and an equivalent number of absence points was randomly selected from all 'absence' points; (ii) creating a training data set and a test data set: the balanced subset of data was randomly divided into two subsets: one for building the models (training set, with 75% of the points) and the other for evaluating the models (test set, with the remaining 25% of the points); (iii) model development: a logistic regression and a BRT model were built using the training set, and parameters of both models were stored; (iv) model evaluation: the model equations from the logistic regression and BRT models were used to generate predictions using the test set, which in turn were assessed using ROC curves, areas under the curve and Cohen's kappa statistic; (v) risk maps: maps of the predictions produced by each model were stored. Steps (i) to (v) were repeated 50 times, and the mean and standard deviation of all statistics and predicted spatial distributions were estimated.

Due to percentage of post-vaccination seropositivity only being available at province level, using yearly data for the period 2007-2009 as unit of analysis, a separate analysis was conducted to quantify the statistical association between antibody response to vaccination expressed as a percentage and two variables: the presence/absence of HPAI H5N1 outbreak records in the province and detection of H5N1 HPAIV detected through national risk-based surveillance activities conducted by the MoA. The post-vaccination seropositivity was analysed as the response variable of a two-way ANOVA with the presence/absence of HPAI H5N1 outbreak records in the province and the year as two explanatory factors. The same analysis was carried out with the presence/absence of H5N1 HPAIV detected through national risk-based surveillance activities and year as explanatory factors. This allowed separating the effect of HPAI H5N1 status from the possible effect of time.

Results

The distribution of HPAI H5N1 outbreaks and HPAIV H5N1 surveillance positives are shown in Fig. 1. Overall, the two analysis



Figure 1. Distribution of HPAI H5N1 outbreaks (grey square) and HPAI H5N1 positive samples identified through surveillance (black triangles) in China. doi:10.1371/journal.ppat.1001308.g001

techniques, logistic regression and BRT, provided consistent results in terms of risk factors being identified. In contrast, the set of risk factors and their effect differed strongly between the outcomes of reported HPAI H5N1 outbreak and risk-based surveillance data.

Based on the logistic regression results, HPAI H5N1 outbreaks were found to be positively associated with human population density and negatively with elevation (Table 2). The BRT models also identified chicken density to be an important variable for discriminating between locations with and without reported HPAI H5N1 outbreaks (BRT weights, Table 2). The averaged BRT model fitted functions shown in Fig. 2 allow a detailed description of these relationships (maps of the predictions coefficient of variation are presented in Figure S1 in Text S1). The predicted risk of HPAI H5N1 outbreak occurrence appears to be constant for densities of chickens ranging from 0 to 10,000 heads/km², then increases to a maximum risk at around 100,000 heads/km². The predicted risk also increases significantly with human population density, starting from a density of 1,000 people/km². We also identify a strong negative relationship with elevation, with the predicted risk function showing two levels, a high risk for elevation ranging from 0-100 m, and a low risk for higher elevations. The predicted risk is relatively flat for all ranges of domestic waterfowl density and percentage of land with surface water.

In contrast, HPAIV H5N1 surveillance positives were found to be positively associated with the density of domestic waterfowl, with percentage of land occupied by water and to human population density (though this factor was not important in the BRT models), and negatively associated with chicken density (Table 2). Here again, the predicted risk function of the BRT models allows a detailed description of these relationships (Fig. 2). The predicted risk of HPAIV H5N1 surveillance positives is constant for waterfowl density ranging between 0 and 10,000 heads/km², then rises sharply for increasing densities. A similar profile as for the HPAI H5N1 outbreak data is found for the association with human population density, with predicted risk increasing with human population density from a density of 1,000 people/km². The predicted risk increases with percentage of area covered by surface water up to a value of approximately 7%, and then remains constant for higher values. The profiles of predicted risk as a function of chicken density and elevation are relatively constant.

The accuracy metrics of the predictions produced by the logistic regression and BRT models are good to excellent, with mean AUC values estimated using the evaluation dataset ranging from 0.864 to 0.967 (Fig. 3). One can note that, as expected, the AUC estimated based on the training data is always better than that estimated using the evaluation dataset, and that this difference is much higher for the BRT models. However, even when assessed using the evaluation dataset, the accuracy of BRT models appears better than that of the logistic regression models. In addition, the accuracy metrics are higher for the models for HPAIV H5N1 risk-based surveillance data than those obtained for the HPAI H5N1 outbreak data. One can note that considering only eastern China in the evaluation of AUC values slightly reduces it's value, but to a marginal extent, showing that the good predictive power does not result from predicting risk over wide desert areas unsuitable to disease spread.

The predicted geographical distribution of HPAI H5N1 presence also differs according to the type of training data (clinical disease outbreaks vs. risk-based surveillance; Fig. 4). Maps generated based on the outbreak data place more emphasis than those based on riskbased surveillance data on north-eastern regions where chicken densities are higher. We also note a marked difference for the outbreak data between the outputs of the logistic regression model and of the BRT, the latter predicting many more clustered areas with high probability of HPAI H5N1 presence. In contrast, the

 Table 2. Results of the bootstrapped logistic regression model and boosted regression trees applied to clinical HPAI H5N1 disease

 outbreak and HPAIV H5N1 serological surveillance data.

					Boosted regression		
	Logistic regressio	n			tree		
Variable	Coef (mean)	Coef. (SD)	Ch. Dev. Rem.	p value	Weight (mean)	Weight (SD)	
Outbreak data (r	n pos. = 184)						
Constant	-2.300	1.731					
ChDnLg	0.288	0.340	0.76	n.s.	23.53	9.15	
DuGeDnLg	-0.324	0.332	0.80	n.s.	7.22	2.93	
HpDnLg	1.544	0.446	31.10	<0.001	28.66	8.54	
DemLg	-1.187	0.346	15.46	<0.001	21.08	9.05	
WaPc	2.268	7.058	0.85	n.s.	6.71	2.95	
RiPc	0.233	1.281	0.42	n.s.	3.26	1.93	
Cint	-1.256	0.910	3.41	0.0650	9.54	3.33	
Surveillance data	a (n pos.=86)						
Constant	-11.404	4.565					
ChDnLg	-1.862	0.835	7.04	0.00797	3.62	2.01	
DuGeDnLg	2.120	1.112	10.57	0.00115	36.86	17.21	
HpDnLg	2.464	0.701	17.81	<0.001	29.01	15.97	
DemLg	0.394	0.813	0.78	n.s.	3.72	2.39	
WaPc	20.802	23.022	5.08	0.0242	18.76	12.14	
RiPc	-0.571	2.437	0.45	n.s.	2.71	1.57	
Cint	-0.434	1.656	0.44	n.s.	5.31	2.22	

doi:10.1371/journal.ppat.1001308.t002



Figure 2. Relationship between risk factors and HPAI H5N1 risk function. The HPAI H5N1 risk functions of the BRT models is plotted as a function of chicken density (ChDnLg, log10 scale), domestic waterfowl density (DuGeDnLg, log10 scale), human population density (HpDnLg, log10 scale), elevation (DemLg, log10 scale), and percentage of land covered by water (WaPc) based on HPAI H5N1 clinical disease outbreak (five first plots) and HPAIV H5N1 risk-based surveillance data (five last plots). The grey lines present the predicted line for each bootstrap, and the black line is the average across all bootstraps.

doi:10.1371/journal.ppat.1001308.g002

distribution of predicted HPAIV H5N1 presence based on riskbased surveillance data identifies areas at risk much more concentrated in the southern part of the country, with outputs from the logistic and BRT models showing similar patterns.

In the analysis at province level, we found that the proportion of seropositivity in post-vaccination surveys was lower in provinces that had reported HPAI H5N1 outbreaks than in those that did not (Fig. 5 left; two-way ANOVA with both year and HPAI H5N1 outbreak status as factor variables; HPAI H5N1 outbreak status: $F_{1,87} = 18.53$, p<0.001; Year: $F_{1,87} = 0.51$, n.s.; interaction term - Year by HPAI H5N1 outbreak status: $F_{1,87} = 0.0264$, n.s.), and in provinces where HPAIV H5N1 had been detected during active surveillance and than in those where this had not been the case (Fig. 5 right; two-way ANOVA with both year and HPAIV H5N1 risk-based surveillance status as factor variables; HPAIV H5N1 risk-based surveillance status: $F_{1,87} = 5.09$, p = 0.026; Year: $F_{1,87} = 2.02$, n.s.; interaction term year by HPAIV H5N1 risk-based surveillance status: $F_{1,87} = 0.172$, n.s.).

Discussion

In China, where approximately 15 billion head of poultry are produced annually with a standing population of 5.6 billion chickens, 760 million ducks and 300 million geese, major regional differences are apparent in ecological systems, husbandry practices, cultural behaviors and economic development with a consequential impact on the distribution of infectious diseases including HPAI H5N1, as well as their maintenance and spread and therefore on disease control options.

To date, spatial studies aiming at identifying HPAI H5N1 risk factors have been undertaken in many countries where the disease was introduced such as Thailand and Vietnam [5,6,14,8,7,19], Korea [20], India and Bangladesh [9,11,21], Romania [22,23] or Africa [24,25]. Only three studies analyzed the distribution of HPAI H5N1 outbreaks in China [4,26]. Of these, only the study by Fang et al. [4] attempts to map the distribution of HPAI H5N1 risk. Whilst highly valuable given that it is the first analysis, the output predicts areas at high risk in ecological areas that would not support the maintenance



Figure 3. ROC curves of the predicted risk of HPAI H5N1 presence/absence. Left and right plots are the ROC curves of the bootstrapped logistic regression models and BRT models, respectively. Top and bottom plots are the ROC curves of the models based on HPAI H5N1 clinical disease outbreak data, and HPAIV H5N1 risk-based surveillance data, respectively. doi:10.1371/journal.ppat.1001308.g003

PLoS Pathogens | www.plospathogens.org



Figure 4. Predicted distribution of HPAI H5N1 risk. Predictions are displayed according to the bootstrapped logistic regression model (left) and boosted regression trees (right), based on reported HPAI H5N1 clinical disease outbreak data (top), or HPAIV H5N1 risk-based surveillance data (bottom). doi:10.1371/journal.ppat.1001308.g004

and transmission of the virus such as in the extremely large desert regions of Inner Mongolia, Tibet and Xinjiang autonomous regions.

In our study, we reported different results for the analyses based on outbreak and risk-based surveillance data. The distribution of reported HPAI H5N1 outbreaks was found to be primarily associated with lowland regions with high human population and chicken density. In contrast, HPAIV H5N1 presence detected through risk-based surveillance activities was found to be associated with regions with high waterfowl densities and were covered by high proportions of surface water. This result is very interesting since it may be a reflection of differences in HPAIV H5N1 pathogenicity between chickens and ducks, combined with environmental and host population conditions supporting virus spread and clinical disease outbreak occurrence as distinct from clinically silent virus persistence.



Figure 5. Province-level percentage of post-vaccination sero-positivity as a function of disease/infection status. The left and right plots are based on reported HPAI H5N1 clinical disease outbreak data and HPAIV H5N1 risk-based surveillance data, respectively. doi:10.1371/journal.ppat.1001308.g005

HPAI H5N1 is far more pathogenic in chickens than in ducks [27,28], though there is also evidence of significant variability in virulence at the species level [29]. In the absence of control or prevention measures, the spread of HPAIV H5N1 and occurrence of clinical disease outbreaks is facilitated in regions where the density of chickens is particularly high, especially in intensive and industrial conditions where high numbers of animals are together facilitating transmission. Such regions are encountered in the north-eastern part of China, where the low cost of grain feed production and a fast-rising demand for poultry meat has supported the rapid development of intensive chicken production. Those intensive poultry production systems invest significant resources in disease prevention measures, and will apply massvaccination of their flocks, thereby preventing HPAIV H5N1 spread within and between farms. However, it is likely given the exceptionally high density of chickens and farms that occasional, albeit rare, lapses in vaccination coverage result in a small number of outbreaks. The human population density risk factor can be interpreted as a proxy of several epidemiological processes that are more likely to occur in highly-populated areas, such as a higher likelihood of outbreak detection and higher possibilities of HPAIV H5N1 transmission through trade and farming-related activities.

In contrast, long-term persistence of HPAIV H5N1 can only be possible if the virus can circulate without being detected or reported. Domestic ducks have been shown to be able to excrete large amounts of virus whilst remaining apparently healthy [27]. Regions rich in domestic waterfowl are hence more prone to long-term persistence of HPAIV H5N1. This can be further exacerbated in geographical areas with an abundance of surface water. Permanent water bodies, rivers, rice paddy fields and canals are the habitat of wild and domestic ducks. One may speculate that water facilitates the transmission between hosts without direct contact through the faecal excretion of the virus, its persistence in the water, and the oral infection of other susceptible hosts sharing the same pond or downstream canal or river. The results indicating that HPAIV H5N1 presence detected through risk-based surveillance is associated with areas that have high waterfowl densities and a high proportion of surface water allows thus a straightforward interpretation. Associations between HPAIV H5N1 and domestic duck density had already been identified in other countries [5,7]. However, no difference between outbreaks and clinically-silent infections was made in these earlier studies, which indeed becomes essential when analyzing HPAIV H5N1 distribution in the context of mass-vaccination such as in China.

Interestingly, the farming and cultural practices encountered in these regions were already described by Shortridge 28 years ago as an avian influenza breeding ground [30]. Among others, Southern China still hosts a massive duck population raised on ponds and rice fields, facilitating frequent faecal-oral transmission of multiple influenza subtypes leading to a year-round and inter-epidemic occurrence of influenza viruses. Historically, agricultural practices in China have developed from the need to feed the people as efficiently as possible, using all available resources, and with little recourse to modern farming methods. Domestic ducks were first moved from rivers to cultivated rice fields at the start of the Qing dynasty in the middle of the 17th century [31,32] to help protect the growing rice from pests. This practice reduces farmers' dependence on chemical insecticides, herbicides, fertilizers and mechanical farming aids and provides a close association between bird, water, rice and people. Ducks raised in ponds are also an important feature in the villages and communities of China, especially in southern China and coastal areas including the waterways of the Pearl River delta which are ideal for rice and fish farming [30]. Furthermore, southern China has always been the focus of influenza experts' attention and often been referred as a hypothetical epicenter of AI pandemic strains. The foundation of this concept was originally raised by Webster et al. [33] and supported by the wide variety of influenza virus subtypes discovered in Southern China during decades [34–37]. More specifically, the distribution of HPAIV H5N1 risk of persistence inferred from the risk-based virological surveillance data and using the logistic and BRT models is similar and highlights different levels of risk according to the following ecological regions (Fig. 4, bottom; see Figure S2 in Text S1 for a map of the zones):

Zone I) In Southern China a large potential zone of virus persistence extends from south of the Yangtze River. This area hosts the vast majority of the Chinese duck population comprising birds for meat or egg production. This area can be subdivided into three areas: I-a) an area which extends from the provinces of Jiangsu, Anhui, Hubei, Jiangxi, Hunan, Guangxi autonomous region down to Guangdong province. This might be one of the most important ecological zones where key epidemiological drivers for emergence, persistence and spread are present, including a huge reservoir population, traditional farming system, a high animal and human population density, some major wild bird congregation sites such as the Poyang lake located in Jiangxi province and an important North – South gradient of poultry trade which crosses this region. This supports the hypothesis of a wider and slightly displaced epicenter of influenza viruses, not only concentrated around the Pearl River delta in Guangdong province but extending south of the Yangtze River and including provinces such as Jiangxi where internal segments of the 1996 geese HPAI H5N1 virus may have originated [38]. I-b) A coastal area stretching from Jiangsu to Guangdong provinces with a risk hotspot in Guangdong province along the Pearl River delta. This strip of coastal land also hosts the typical duck pond system where the risk of infection and disease is present. I-c) Few isolated areas within this geographical zone displaying an increased risk located in Yunnan, Guangxi autonomous region, Guizhou, Sichuan and Chongqing provinces which have experienced either outbreaks in the past (Guangxi autonomous region and Yunnan) or only reported viral circulation (Sichuan and Chongqing provinces).

Zone II) A vast geographical area in the West and North, displaying radically different geography, socio-economic and animal production features and characterized by scattered and isolated spots of higher predicted risk. This includes specifically southern Tibet autonomous region and scattered areas in the North and South of Xinjiang autonomous region where sporadic outbreaks have occurred in the past.

Zone III) In the North-East of the Yangtze River, a region where the contribution to disease persistence seems fairly limited while localized areas at higher risk of outbreaks encroach regions of intensive production where the disease could rapidly spread in case of virus introduction and breach in biosecurity. This region extends from Shangdong into Liaoning, Jilin and Helongjiang provinces. These provinces are characterized by denser human population and large-scale commercial poultry production, and were predicted as high risk based on the reported clinical disease outbreak data (Fig. 4 top). In these regions of North-Eastern China, chicken production and marketing systems are intensifying and concentrating in response to economic growth and urbanization. Substantial numbers of poultry are now processed at large-scale slaughterhouses in this region, while the majority of poultry are still sold through live poultry markets in the South of the country. In the colder north-eastern provinces water birds are also housed and kept more intensively.

The apparent persistence of HPAIV H5N1 in those regions has two main implications. First, given the possible presence of silent infection involving an extremely high population of domestic waterfowl, eradication of the virus through massive vaccination appears extremely difficult, although it has successfully reduced the number of outbreaks. Vaccination has been one major component of the government policy to curb the spread of the disease and reduce the incidence of outbreaks of clinical disease and of transmission of infection. China uses more vaccine against avian influenza than any other country and Chinese veterinary authorities base much of their control and preventive strategy around vaccination [39-41]. More than 13 billion doses of AI vaccine have been used each year since 2007 [39] and the objectives of the national strategy are to reach a 100% vaccination coverage for the national poultry population and to ensure an effective immune response (defined as sero-conversion in bird with titres >4Log2 when measured by HI test) in more than 70% of the nationwide poultry population all year round. In this study, we also analyzed the post-vaccination surveillance data collected at provincial level since January 2007 and found that provinces where clinical HPAI H5N1 outbreaks had been reported or HPAIV H5N1 detected had a lower level of post-vaccination seropositivity, confirming that increased protection does indeed result in lower disease outbreak or infection risks but would require an approach better targeted at identified high risk areas to drastically reduce the viral load in the environment.

Second, the different regions of China are not independent and are possibly epidemiologically linked through poultry trade and likely also through wild bird migrations. High production-demand discrepancies lead to long-distance trade of poultry products (e.g. chicken from the north exported to southern provinces, or duck meat exported from the south to the north). In addition, areas such as the Poyang lake, where a large population of domestic waterfowl is raised in close proximity to thousands of overwintering wild waterfowl, could favour the transmission between wild and domestic waterfowl and lead to long-distance transmission of the virus. As a consequence, the persistence of HPAIV H5N1 in some particular regions may influence the chances of introduction into other more distant regions. For instance, the wild bird 2.2 clade which was associated with the origin of the Qinghai lake epidemic in 2005 in West China was responsible for a major outbreak during the same year in domestic poultry in Liaoning province, in the north eastern part of the country. Likewise, the 2006 Shanxi strain also grouped into the clade 7 cluster present in North-Central China has been found in Jiangsu province in South Eastern China. There is a complex pattern of links that exists between these different ecological regions that offers hiatuses for viruses to escape their reservoir areas and invade others.

Continued efforts pursued by the Ministry of Agriculture, its affiliated research centers and local veterinary authorities to strengthen the HPAI national surveillance program and its control strategy have resulted in a steady decrease in the number of outbreaks reported since 2004 and a better understanding of HPAIV H5N1 infection distribution in space, time and within traditional marketing systems known as live bird markets.

However, national surveillance programs have also demonstrated that HPAIV H5N1 continues to circulate in poultry on a regular basis. Since 2007, an apparent increase in virus detection is

References

- Xu X, Subbarao K, Cox NJ, Guo Y (1999) Genetic Characterization of the Pathogenic Influenza A/Goose/Guangdong/1/96 (H5N1) Virus: Similarity of Its Hemagglutinin Gene to Those of H5N1 Viruses from the 1997 Outbreaks in Hong Kong. Virology 261: 15–19.
- Li KS, Guan Y, Wang J, Smith GJD, Xu KM, et al. (2004) Genesis of a highly pathogenic and potentially pandemic H5N1 influenza virus in eastern Asia. Nature 430: 209–213.
- Shortridge KF (1995) The next pandemic influenza virus? The Lancet 346: 1210–1212.

believed to represent the result of increased and intensive efforts made by the Ministry of Agriculture to detect the virus through targeted risk-based surveillance activities at live bird markets and high-risk farms in a context of massive vaccination efforts that could potentially mask the clinical expression of the disease within a large population of immunized birds.

Although the epidemiology of HPAIV H5N1 in China does not seem to present radically different features compared with neighboring countries also affected by the disease, it remains unique in terms of the abundance of reservoir species both domestic or wild, providing ample opportunities for a sustained and rapid evolution of the virus and requiring intensive virus monitoring for pandemic preparedness matters. While revisiting the concept of epicenter for pandemic strains of avian origin, the results of this study represent major improvements over previous efforts in mapping the risk of HPAI H5N1 in two main aspects.

First, it allows identifying several risk factors of animal, environmental and anthropogenic nature, with clear biological and epidemiological interpretation. Second, the bootstrapped statistical modeling allows us to robustly estimate the predictive power of our model, but also to map the uncertainty that goes with our predictions (Figure S1 in Text S1), which is useful information for an applied use of these maps.

Combining innovative modeling techniques with data of improved quality and integrating measures of infection persistence, our results have broad fundamental implications in a country where understanding of the ecology of influenza viruses, although of utmost importance for pandemic preparedness purposes, has remained until now mostly speculative.

Finally, the analyses presented in this study may be improved in the future by several complementary approaches. First, the potential transmission through trade patterns and bird migration should be more comprehensively assessed. An increasing amount of data are being collected on both aspects, and this will ultimately contribute to better understanding of how areas of high potential for HPAIV H5N1 persistence may be connected to each others. Second, the results could be further integrated into an Asia-wide improved understanding of HPAIV H5N1 distribution models, benefiting from several studies that have been undertaken in neighboring countries. Third, information on true negatives obtained through the national surveillance programme would reduce the risk of including false-negatives in the analyses, and provide higher resolution estimates of the relative importance of risk factors.

Supporting Information

Text S1 Supplementary information Figure S1 and Figure S2. Found at: doi:10.1371/journal.ppat.1001308.s001 (0.61 MB DOC)

Author Contributions

Conceived and designed the experiments: VM. Performed the experiments: VM. Analyzed the data: VM DUP XZ XX DJP MG. Contributed reagents/materials/analysis tools: FG. Wrote the paper: VM DUP MG.

- Fang LQ, de Vlas SJ, Liang S, Looman CW, Gong P, et al. (2008) Environmental factors contributing to the spread of H5N1 avian influenza in mainland China. PLoS ONE 3: e2268.
- Gilbert M, Chaitaweesub P, Parakarnawongsa T, Premashthira S, Tiensin T, et al. (2006) Free-grazing ducks and highly pathogenic avian influenza, Thailand. Emerg Infect Dis 12: 227–234.
- Gilbert M, Xiao X, Pfeiffer DU, Epprecht M, Boles S, et al. (2008) Mapping H5N1 highly pathogenic avian influenza risk in Southeast Asia. Proc Natl Acad Sci USA 105: 4769–4774.

- Pfeiffer DU, Minh PQ, Martin V, Epprecht M, Otte MJ (2007) An analysis of the spatial and temporal patterns of highly pathogenic avian influenza occurrence in Vietnam using national surveillance data. Vet J 174: 302–309.
- Tiensin T, Ahmed SSU, Rojanasthien S, Songserm T, Ratanakorn P, et al. (2009) Ecologic Risk Factor Investigation of Clusters of Avian Influenza A (H5N1) Virus Infection in Thailand. J Infect Dis 199: 1735–1743.
- Loth L, Gilbert M, Osmani M, Kalam M, Xiao X (2010) Risk Factors and Clusters of Highly Pathogenic Avian Influenza H5N1 in Bangladesh. Prev Vet Med 96: 104–113.
- Paul M, Tavornpanich S, Abrial D, Gasqui P, Charras-Garrido M, et al. (2010) Anthropogenic factors and the risk of highly pathogenic avian influenza H5N1: prospects from a spatial-based model. Vet Res 41: 14.
- Biswas PK, Christensen JP, Ahmed SSU, Das A, Rahman MH, et al. (2009) Risk for Infection with Highly Pathogenic Avian Influenza Virus (H5N1) in Backyard Chickens, Bangladesh. Emerg Infect Dis 15: 1931–1936.
- Ward MP, Maftei D, Apostu C, Suru A (2008) Environmental and anthropogenic risk factors for highly pathogenic avian influenza subtype H5N1 outbreaks in Romania, 2005–2006. Vet Res Commun 32: 627–634.
- Williams RAJ, Peterson AT (2009) Ecology and geography of avian influenza (HPAI H5N1) transmission in the Middle East and northeastern Africa. Int J Health Geogr 8: 47.
- Henning J, Pfeiffer DU, Vu LT (2008) Risk factors and characteristics of H5N1 Highly Pathogenic Avian Influenza (HPAI) post-vaccination outbreaks. Vet Res 40: 15.
- Elith J, Leathwick JR, Hastie T (2008) A working guide to boosted regression trees. J Anim Ecol 77: 802–813.
- Elith J, Graham CH, Anderson RP, Dudík M, Ferrier S, et al. (2006) Novel methods improve prediction of species' distributions from occurrence data. Ecography 29: 129–151.
- McPherson JM, Jetz W, Rogers DJ (2004) The effects of species' range sizes on the accuracy of distribution models: ecological phenomenon or statistical artefact? J Appl Ecol 41: 811–823.
- Marmion M, Luoto M, Heikkinen RK, Thuiller W (2009) The performance of state-of-the-art modelling techniques depends on geographical distribution of species. Ecol Model 220: 3512–3520.
- Minh PQ, Morris RS, Schauer B, Stevenson M, Benschop J, et al. (2009) Spatiotemporal epidemiology of highly pathogenic avian influenza outbreaks in the two deltas of Vietnam during 2003–2007. Prev Vet Med 89: 16–24.
- Yoon H, Park CK, Nam HM, Wee SH (2005) Virus spread pattern within infected chicken farms using regression model: the 2003–2004 HPAI epidemic in the Republic of Korea. J Vet Med B 52: 428–431.
- Adhikari D, Chettri A, Barik SK (2009) Modelling the ecology and distribution of highly pathogenic avian influenza (H5N1) in the Indian subcontinent. Cur Sci India 97: 73–78.
- Ward MP, Maftei DN, Apostu CL, Suru AR (2009) Association Between Outbreaks of Highly Pathogenic Avian Influenza Subtype H5N1 and Migratory Waterfowl (Family Anatidae) Populations. Zoonoses Public Health 56: 1–9.
- Ward MP, Maftei D, Apostu C, Suru A (2008) Geostatistical visualisation and spatial statistics for evaluation of the dispersion of epidemic highly pathogenic avian influenza subtype H5N1. Vet Res 39: 22.
- Cecchi G, Ilemobade A, Le Brun Y, Hogerwerf L, Slingenbergh J (2008) Agroecological features of the introduction and spread of the highly pathogenic avian influenza (HPAI) H5N1 in northern Nigeria. Geospatial Health 3: 7–16.
- Williams RAJ, Fasina FO, Peterson AT (2008) Predictable ecology and geography of avian influenza (H5N1) transmission in Nigeria and West Africa. T Roy Soc Top Med H 102: 471–479.

- Oyana TJ, Dai D, Scott KE (2006) Spatiotemporal distributions of reported cases of the avian influenza H5N1 (bird flu) in Southern China in early 2004. Avian Dis 50: 508–515.
- Hulse-Post DJ, Sturm-Ramirez KM, Humberd J, Seiler P, Govorkova EA, et al. (2005) Role of domestic ducks in the propagation and biological evolution of highly pathogenic H5N1 influenza viruses in Asia. Proc Natl Acad Sci USA 102: 10682–10687.
- Kim J, Negovetich NJ, Forrest HL, Webster RG (2009) Ducks: the "Trojan horses" of H5N1 influenza. Influenza Other Respi. Viruses 3: 121–128.
- Brown JD, Stallknecht DE, Valeika S, Swayne DE (2007) Susceptibility of wood ducks to H5N1 Highly pathogenic avian influenza virus. J Wildl Dis 43: 660–667.
- Shortridge KF (1982) Avian influenza A viruses of southern China and Hong Kong: ecological aspects and implications for man. Bull World Health Organ 60: 129–135.
- Shortridge KF (2005) Avian influenza viruses in Hong Kong: zoonotic considerations. Wageningen UR Frontis Series 8: 9.
- Wallace RG (2009) Breeding Influenza: The Political Virology of Offshore Farming. Antipode 41: 916–951.
- Webster RG, Campbell CH, Granoff A (1973) The "in vivo" production of "new" influenza viruses: III. Isolation of recombinant influenza viruses under simulated conditions of natural transmission. Virology 51: 149–162.
- Chang WK (1969) National influenza experience in Hong Kong, 1968. Bull World Health Organ 41: 349–351.
- Cheung CL, Vijaykrishna D, Smith GJD, Fan XH, Zhang JX, et al. (2007) Establishment of influenza A virus (H6N1) in minor poultry in southern China. J Virol 81: 10402–10412.
- 36. Shortridge K (1982) An influenza epicenter ? The Lancet 320: 812-813.
- Xu KM, Smith GJD, Bahl J, Duan L, Tai H, et al. (2007) The genesis and evolution of H9N2 influenza viruses in poultry from southern China, 2000 to 2005. J Virol 81: 10389–10401.
- Mukhar MM, Rasool ST, Song D, Zhu C, Hao Q, et al. (2007) Origin of highly pathogenic H5N1 avian influenza virus in China and genetic characterization of donor and recipient viruses. J Gen Virol 88: 3094–3099.
- Chen H (2009) Avian influenza vaccination: the experience in China. Rev Sci Tech 28: 267–274.
- Chen H (2009) H5N1 avian influenza in China. Sci China, C, Life Sci 52: 419–427.
- Chen H, Bu Z (2009) Development and application of avian influenza vaccines in China. Curr Top Microbiol Immunol 333: 153–162.
- Robinson TP, Franceschini G, Wint W (2007) The Food and Agriculture Organization's Gridded Livestock of the World. Vet Ital 43: 745–751.
- 43. CIÈSIN, IPFRI, CIAT (2005) Global Rural-Urban Mapping Project (GRUMP), Alpha Version. Center for International Earth Science Information Network (CIESIN), Columbia University; International Food Policy Research Institute (IPFRI); The World Bank; Centro Internacional de Agricultura Tropical (CIAT).
- 44. Jiyuan L, Mingliang L, Xiangzheng D, Dafang Z, Zengxiang Z, et al. (2002) The land use and land cover change database and its relative studies in China. J Geogr Sci 12: 275–282.
- Xiao X, Boles S, Liu J, Zhuang D, Frolking S, et al. (2005) Mapping paddy rice agriculture in southern China using multi-temporal MODIS images. Remote Sens Environ 95: 480–492.
- LDAAC (2004) Global 30 Arc-Second Elevation Data Set GTOPO30 Land Process Distributed Active Archive Center.